Author's response to reviews

Title: Use of near-infrared light to reduce symptoms associated with Restless Legs Syndrome: a case report

Authors:

Ulrike H Mitchell (rike_mitchell@byu.edu)

Version: 5 Date: 24 June 2010

Author's response to reviews: see over
Enclosed is the proof-read version of my case report, titled “Use of near-infrared light to reduce symptoms associated with Restless Legs Syndrome: a case report”.

I made the following changes as requested:

- Removed qualifications and job position from the author list
- Changed the affiliation format
- Changed the color of the manuscript to black
- Removed the running header
- Deleted the figure from the main manuscript

Thank you again for your time and consideration!

Sincerely,

Ulrike H. Mitchell